Author's response to reviews

Title: Impact of maternal metabolic abnormalities in pregnancy on human milk and subsequent infant metabolic development: methodology and design

Authors:

Sylvia H Ley (s.ley@utoronto.ca)
Deborah L O'Connor (deborah_l.oconnor@sickkids.ca)
Ravi Retnakaran (rretnakaran@mtsinai.on.ca)
Jill K Hamilton (jill.hamilton@sickkids.ca)
Mathew Sermer (msermer@mtsinai.on.ca)
Bernard Zinman (zinman@lunenfeld.ca)
Anthony J Hanley (anthony.hanley@utoronto.ca)

Version: 3 Date: 5 October 2010

Author's response to reviews:

Dear Editor-in-Chief Melissa Norton:

Please find attached the revised manuscript entitled, "Impact of maternal metabolic abnormalities in pregnancy on human milk and subsequent infant metabolic development: methodology and design". We have made small typographical changes to the main document: "t" was inserted into the word "overweight" as it was missing in the previous version page 11, and typographical errors were corrected in references #10, 11, 61, 63, 72, and 74.

Please do not hesitate to contact me for any additional information.

Best regards,
A. Hanley
Corresponding author
Department of Nutritional Sciences,
University of Toronto, FitzGerald Building,
150 College Street, Room 341, Toronto, Ontario, Canada M5S 3E2.
Telephone: 1 416 978 3616
Fax: 1 416 978 5882
E-mail: anthony.hanley@utoronto.ca